ARTICLE





Urban Transition Toward Environmental Sustainability: Instrumentation and Institutionalization of Co-Creation

Ben Vermeulen¹⁰, Lennart Winkeler²⁰, and Mohar Kalra³

¹ United Nations University – Maastricht Economic and Social Research Institute on Innovation and Technology (UNU-MERIT), The Netherlands

² Research Group for Resilient Energy Systems, University of Bremen, Germany

³ University of the Arts Bremen, Germany

Correspondence: Ben Vermeulen (ben.vermeulen@maastrichtuniversity.nl)

Submitted: 18 November 2024 Accepted: 27 April 2025 Published: 29 July 2025

Issue: This article is part of the issue "Co-Creation With Emerging Technologies to Address Climate Challenges in Cities" edited by Cesar Casiano Flores (University of Twente), A. Paula Rodriguez Müller (European Commission, Joint Research Centre), and Evrim Tan (University of Munster), fully open access at https://doi.org/10.17645/up.i439

Abstract

The transition in cities toward environmental sustainability requires transforming urban subsystems such as energy, transport, and waste infrastructure. Based on the frameworks of strategic spatial planning (SSP) and urban transition management (UTM), the urban transition is conceptualized as a long-term process in which stakeholders co-create a vision and a strategic plan, which is subsequently implemented in multiple relatively short-term projects transforming these urban subsystems. While co-creation is emerging in urban planning, ambiguity remains regarding the development and use of co-creation instruments in transforming urban subsystems. This article therefore has two aims: first, to develop a typology of co-creation instruments for urban transition planning and management; and second, to examine the institutionalization of their development and use. The article follows an iterative inductive-deductive search method to make an inventory of instruments, after which four main types are identified: participatory planning and communication tools, expert planning support systems, urban living labs, and virtual transformation labs. Several challenges in using these instruments are identified, including the need to acquire governance and digital skills, and to keep tools and data up to date. This article subsequently examines the capabilities that need to be institutionalized to support the use and development of these instruments across multiple projects. Capabilities needed are stakeholder engagement and collaborative governance, the participatory design and updating of digital tools, maintenance of urban subsystem and city development models, definition of transition scenarios and experiments, and interpretation of (simulation) results. Additional capabilities are needed to manage the project portfolio and facilitate learning within and across projects. Ultimately, a "Transition Planning Office" is proposed to institutionalize these capabilities and, by doing so, to complement UTM's focus on independent vision and agenda formulation with sustained involvement in



long-term planning, and to support SSP's call for more strategic urban planning through project portfolio management and instrument use and development.

Keywords

co-creation; environmental sustainability; institutionalization; instrumentation; living lab; planning support system; strategic spatial planning; urban transformation lab; urban transition management

1. Introduction

Many cities around the world are engaged in the transition toward environmental sustainability (see e.g., the Global Covenant of Mayors), notably aiming for low greenhouse gas emissions, low energy consumption, clean transport options, adequate waste management, green building practices, and so forth. This urban transition is a long-term, complex process fraught with uncertainty in which urban subsystems such as infrastructure (e.g., energy, transport, waste), the built environment, and the green-blue ecological network are to be transformed to meet sustainability targets. However, most of these urban subsystems rely on costly infrastructure, capital goods, and complementary technologies, with entrenched use and support by institutions, which together cause a "carbon lock-in" (see e.g., Seto et al., 2016). Incremental innovations of the existing system generally do not suffice, and instead, a radical change of the system is required (Loorbach, 2022). Mitigation of and adaptation to climate change need to be integrated into urban design, land use, and urban planning praxis (Bai et al., 2021; Raven et al., 2018). This may also require changes to regulations, institutions, and public services to promote sustainable consumption and economic activities (Keivani, 2009; Talen, 2012). This radical change needs to be proactively governed (Wittmayer & Loorbach, 2016) by engaging the diverse set of stakeholders (e.g., urban planners, policymakers, public utility providers, local communities, construction companies) to co-create solutions to local particularities and to find trade-offs between conflicting interests. This resonates with planning theorists' calls for progressive participation (Healey, 2020) and transforming traditional planning to facilitate co-creation by engaging a broad range of actors (Albrechts, 2013).

Despite the increasing attention on co-creation, there is still considerable ambiguity (Brandsen & Honingh, 2018) regarding its operationalization in urban transitions. Contributing to this is the fragmented evolution of conceptual frameworks, the rapid emergence of a variety of instruments, and a lack of clarity on the role of urban planners and other stakeholders therein. As a conceptual basis for examining co-creation activities in urban transition, this article synthesizes two conceptual frameworks that extend beyond the traditional master plan and zoning regulations and redefine the role of expert planning professionals in this process. First, strategic spatial planning (SSP; Albrechts, 2004, 2013, 2015) integrates long-term vision formulation and short-term change projects with adaptive learning on both vision and operational project goals. Second, urban transition management (UTM) promotes the establishment of a transition team of change agents to co-create an agenda to develop radically different urban subsystems (Frantzeskaki, 2022; Frantzeskaki et al., 2018; Nevens et al., 2013; Roorda et al., 2014; Wittmayer & Loorbach, 2016). While SSP seeks to reform existing practices and instruments toward strategic, stakeholder-oriented planning, UTM is deliberately placed outside of existing urban planning and does not prescribe methods or instruments for the long-term management of individual projects or the portfolio thereof (see Wolfram, 2018, p. 111). In synthesis, the urban transition is perceived as a long-term process driven by a shared vision for the future city, whereby



this vision is realized through multiple, relatively short-term, co-created projects that transform urban subsystems. The tools, methods, and approaches for the co-creation of the urban transition process continue to evolve with the rise of new planning paradigms, theoretical conceptualizations, methodologies, data availability, and technical advancements. Not only are traditional planning tools being extended with new sustainability criteria and used in participatory planning, but also new instruments, such as living labs, digital twins, and city development simulations, are being developed. Moreover, applications of online platforms for participation, data-intensive analytical tools, and digital decision support systems are on the rise, also in urban transition planning. However, there is currently neither a comprehensive overview of instruments nor a clear understanding of how to adequately develop and use them in practice. As such, the article has two aims: to develop a typology of co-creation instruments and to examine the institutionalization of the development and use of these instruments in urban transition planning and management. First, the typology is developed by making an inventory of instruments according to an iterative inductive-deductive approach (see Mayring, 2000) and subsequently classifying and clustering them into types. Hereby, the instruments are assessed on three dimensions: whether they focus primarily on qualitative information exchange or rather quantitative data analysis using digital tools; whether they are expert-led or rather follow a participatory or collaborative governance approach; and whether they support short-term decision-making or rather long-term strategic planning, experimentation, and learning. Based on this classification, four distinct types of instruments for urban transition are discerned: participatory planning and communication platforms, expert planning support systems, urban living labs, and virtual transformation labs. In addition, the article identifies specific challenges in using digital co-creation tools, particularly in planning support systems and virtual transformation labs. Second, given the long horizon and the multiple projects in which co-creation instruments are to be used, the article examines the institutionalization of their development and use. Urban transition planning requires building capabilities not only for stakeholder engagement for vision formulation and strategic planning, but also for the participatory design, maintenance, and use of urban subsystem and city (development) models, as well as the analysis of (simulation) data. Moreover, given the rapid evolution of co-creation initiatives and instruments, planners need capabilities to monitor and learn within and across projects (possibly in other cities) as well as to manage the overall portfolio of transition projects. This article thus contributes an operational perspective on co-creation instrumentation and institutionalization to the existing literature on urban transition and planning theory. While not a primary aim, the article also proposes a "Transition Planning Office" (TPO) as the institutionalization of these capabilities. The TPO complements the integration of strategic planning proposed by SSP and the deliberately independent transition arena proposed by UTM.

The structure of the remainder of the article is as follows. Section 2 provides the conceptual framework for the co-creation of urban transition and builds upon the SSP and UTM approaches to outline the purpose and challenges of instrumentation and institutionalization. Section 3 presents the typology of instruments, explains the methodology used to develop this typology, and discusses challenges in the implementation and adaptation of these instruments. Section 4 discusses the institutionalization of transition planning and management, focusing on capabilities required for stakeholder governance and using instruments within and across the range of urban subsystem transformation projects. Section 5 provides conclusions and insights into the instrumentation and institutionalization of co-creating transition planning and management.



2. Conceptual Framework

2.1. Co-Creation of the Urban Transition to Environmental Sustainability

As described above, urban subsystems may be in a carbon lock-in (cf. Seto et al., 2016), such that their transformation to become environmentally sustainable needs to be proactively governed (Hölscher, 2018; Loorbach, 2022; Wittmayer & Loorbach, 2016) and may require changes not only to infrastructure, but also to regulations, institutions, and public services (Keivani, 2009; Talen, 2012). However, alternative transformations may exist. For instance, air pollution from transportation can be reduced by introducing congestion pricing during peak hours, implementing ride-sharing systems, enhancing public transportation, restructuring the road network, and making cities walkable. Similarly, CO_2 emissions from energy consumption can be reduced through energy-saving campaigns, subsidizing smart appliances, switching to green electricity in public buildings, promoting the purchase of solar panels, subsidizing building insulation, and imposing stricter building standards. Which options are feasible and preferred may depend on local circumstances, stakeholder capabilities and interests, as well as interrelations between transition solutions.

The diversity of stakeholders (e.g., policymakers, public utility providers, local communities, companies, investors) and their different interests make the transformation of urban subsystems a potential source of conflicts. Breaking out of the carbon lock-in therefore requires the co-creation of transformative solutions with trade-offs that are acceptable to these stakeholders. Arguably, the government has both the mandate and responsibility to engage these stakeholders and facilitate this co-creation process. This is also in line with the broader trend toward greater involvement of stakeholders in governmental decisions in general, and of stakeholder engagement and participatory governance in urban planning in particular (Healey, 2020). Stakeholders are not merely involved for tokenism or legitimation but are genuinely delegated decision-making power (Albrechts, 2013; Fugini et al., 2016; Leino & Puumala, 2021). Co-creation thereby leads to solutions that are neither imposed by expert decision-makers (top-down) nor driven by interest groups, citizens, or powerful stakeholders (bottom-up) but rather by engaging a variety of stakeholders across hierarchical relationships (Leino & Puumala, 2021).

However, in general, the concept and instrumentation of co-creation are still ambiguous and emerging (see e.g., Brandsen & Honingh, 2018). The understanding of what co-creation is (and could be) in the context of the urban transition is rapidly developing with (a) the conceptions of the urban transition in terms of goals, phases, and tasks, (b) the various perspectives on the transition-related activities and the role of various institutions therein (see SSP and UTM), and (c) the variety of operational instruments, methods, and approaches.

As discussed in the following subsections, the urban transition is seen as a long-term process driven by a co-created vision and development agenda, notably featuring relatively short-term projects that transform urban subsystems (infrastructures, utilities, services, built environment) aligned with that vision. As such, co-creation is understood as a governance approach in which an "urban transition agent" collaborates with stakeholders to formulate a shared vision and roadmap for urban developments. However, over time, new challenges and opportunities may arise, and specific developments may need to be deferred or adjusted. Given its inherent complexity and uncertainty, the transition process cannot be tightly planned but instead features experimentation, monitoring, learning, and adaptation both within and across projects. This requires alternative governance structures not only for the initial formulation of the vision and agenda but also to



facilitate the long-term engagement of stakeholders in co-creation activities (see e.g., Mahmoud et al., 2021). Thus, in the context of urban transition, co-creation is de facto characterized by the instruments applied to a range of tasks and the institutionalization of the involvement of stakeholders. The next subsections discuss key conceptual frameworks for planning and managing urban transformation, followed by the outlines of the instrumentation and institutionalization of co-creation within these frameworks.

2.2. SSP for Sustainability

Urban planning has evolved from being an authoritative, regulatory approach with institutionalized, government-led urban development practices to an approach that uses stakeholder engagement, participation instruments, and various governance modes (Faludi, 1973; Healey, 2020). SSP emerged as one of the responses to the dominant expert-led, top-down approaches. As Albrechts (2013, p. 52) summarizes:

Strategic spatial planning is looked upon as a transformative and integrative public sector-led, but co-productive, socio-spatial process through which visions or frames of reference, the justification for coherent actions, and the means for implementation are produced that shape, frame and reframe what a place is and what it might become (Albrechts, 2010: 1117; Motte, 2006; Oosterlynck et al., 2011; Van den Broeck et al., 2010).

SSP revolves around participatory involvement of empowered citizens and a broader set of stakeholders (firms, environmental groups, consultants) and translating these visions into short-term actions (Albrechts, 2004). Moreover, rather than merely soliciting citizens' opinions to subsequently have professionals do the planning and projects, these citizens are involved in "agenda setting, problem formulation, the shaping of the content of policies, plans and projects and the delivery as well" (Albrechts, 2013, p. 53). SSP thereby challenges existing relationships and rather seeks to empower and include innovative change agents in an adaptive, co-creation process (Albrechts, 2013, 2015). In SSP, a shared, long-term vision is co-created and subsequently implemented in short-term projects, adapting these in continuous learning while involving stakeholders to mobilize networks and capabilities.

Urban development has traditionally been (and generally still is) regulated through master plans and zoning regulations, and even though these may, in some cases, cater to sustainability goals (Jepson & Haines, 2014), transforming existing energy and transport infrastructure to meet these goals would typically necessitate radical redevelopment. In practice, urban planning does use methods and systems to reform land use and energy and transportation infrastructure, albeit typically not from a sustainability transition perspective (Walsh, 2018). Generally, the planning of urban subsystems is path-dependent rather than path-breaking (see e.g., Malekpour et al., 2015). Although SSP should be integrated into regular urban planning approaches (Wolfram, 2018, p. 111), it is intended to be complementary to, rather than a substitute for, the common planning tools such as land use, master planning, or zoning.

2.3. UTM

Unlike SSP, UTM explicitly starts from the premise that radical change is needed and that the ideas for this are best formulated in a "safe space" sheltered from the influence of the actors within the unsustainable pre-existing system. It advocates bringing together a deliberately broad variety of "frontrunners" to develop



a shared understanding of the urban transition challenges, formulate a collective, long-term vision for the city, and translate this into flexible pathways with short-term active experimentation and learning (Wittmayer & Loorbach, 2016; Wittmayer et al., 2018). As described in detail in Roorda et al. (2014), an initiating agent kickstarts the formation of a "transition team." This team drives the management process, analyzes the urban subsystems to be transformed, and conducts a stakeholder analysis to invite change agents in a "transition arena." This transition arena is a small, diverse group of change agents from various backgrounds (e.g., municipality, companies, research institutes, citizens) and domains (e.g., energy, transport). Over the course of several meetings, the arena participants formulate the transition challenge, problem framing, and visions for the future. Using these visions, arena participants engage in backcasting methods to define possible pathways to realize the visionary image (such as a roadmap of actionable goals towards transforming the city). Transition experiments are used to explore and assess the viability of radical changes to resolve transition challenges. The action plans and their associated transition experiments used to realize long-term visions and pathways are documented in the transition agenda. This agenda encompasses "a strategic perspective—a transition narrative—that can be used as an anchor point for new initiatives and policy" (Roorda et al., 2014, p. 32). This agenda is (merely) a means to engage the wider audience and influence the agenda of others, e.g., by organizing networking events, by seeking publicity, by integrating elements into existing processes, or by introducing transition concepts and practices within municipal organizations.

Since the agenda merely sets the direction for the long term and guides actions in the short term, the next step would be to initiate transition projects. Such projects hinge on actors that gather the required competences, funding, and resources to operationalize the ideas in the agenda. This is a "working group" of actors responsible for project management and involving companies and policymakers. Moreover, a coordinator is required to facilitate ongoing engagement and reflection among actors involved in the transition process.

Despite the name of the approach, transition management does not prescribe methods or instruments for the *long-term* management of single initiatives, let alone the portfolio of transition projects. Indeed, the transition arena is just a temporary, short-term process at the front-end of the transition, and is foreseen to cease to exist after formulating the agenda and engaging the wider audience. Nonetheless, there is a lack of clarity about the actual institutionalization of the long-term transition activities. In addition, at some point, moving forward with changes to a socio-technical system or public service becomes a matter of design, engineering, project planning, procurement, and execution. The room for adaptation and co-creation tends to diminish. So, it seems that, to a certain extent, short- and mid-term urban planning methods and activities are expected to exist side-by-side with the long-term transition perspective.

2.4. Instrumentation and Institutionalization of Co-Creation in Urban Transitions

Both the long-term urban development vision and agenda as well as the portfolio of relatively short-term projects transforming urban subsystems are co-created. Each project is a transition challenge with a variety of solutions to consider. Currently, urban planners take the long-term master plan, land development plan, and zoning regulation as guideline. So, arguably, urban planners lack a toolkit of instruments to research, develop, and implement transition solutions in actual urban planning practice (cf. van de Ven et al., 2016). In part, this may be due to an inherent tension between long-term, "vision-pulled" transition processes that emphasize experimentation and reflection and the relatively short-term urban planning and design projects that merely



extrapolate from existing developments. Clearly, striving for the materialization of plans is at odds with keeping options open and postponing changes that are costly to reverse. So, this experimentation and analysis occurs prior to the planning, requiring additional instruments and arrangements. Neither core UTM nor SSP literature is explicit about instrumentation for analysis and decision-making. Moreover, core UTM literature refrains from discussing the institutionalization of front-end vision formulation, long-term process management, or short-term practical planning activities.

Despite their commonalities, both SSP and UTM have their own instrumentation and institutionalization requirements. Notably, UTM requires qualitative support for radical innovation, roadmap backcasting with short-term experimentation, monitoring, learning, as well as adopting grassroots initiatives. In contrast, since SSP is integrated into existing planning practices, it may have more use for long-term data and extrapolated trends and means for adjusting existing policy programs and zoning regulations. Moreover, co-creation with stakeholders far beyond regular planning commissions is prominent in both, so instruments and practices need to be developed in any case. Either way, given its novelty, complexity, and scale, the urban planning praxis has to develop "competences for designing transition management processes in cities and engaging with multidisciplinary knowledge in solution-seeking processes" (Frantzeskaki et al., 2018). Urban planning offices need to learn how to set up, develop, plan, and ultimately manage the execution of these transition projects. In the following two sections, both instrumentation and institutionalization for co-created transition are examined in detail.

3. Instrumentation of Transition Co-Creation

Over the past decades, the toolkit of instruments in urban planning has been evolving, notably in response to changes in aspirations, considerations, and responsibilities in planning theory (Allmendinger & Tewdwr-Jones, 2001), the rise and fall of movements (Garde, 2020), and the emergence of urban ideals such as the eco-city, walkable city, 15-minute city, etc. Moreover, new instruments have been developed leveraging digital technology and its modeling and decision-support capabilities, such as the smart city and digital twin city (see e.g., Batty, 2001; Silva, 2010). In addition, there has been a rise in online platforms for citizen participation in general (see Ragi Eis Mendonca & De la Llata, 2023; Rodriguez Müller, 2022) and urban planning in particular (see e.g., Abdalla et al., 2016; Ertiö, 2015; Falco & Kleinhans, 2018; Giannoumis & Joneja, 2022; Hovik et al., 2022). However, as discussed in the previous section, the co-creation of urban transformation requires additional instruments, particularly for at least two distinct process steps. Firstly, there is co-creation in the formulation of visions and the agenda to develop the envisioned city. The latter could be a roadmap of actionable goals and tasks based on backcasting, expert assessments, or analysis with a (computational) city development model. Secondly, there is co-creation of a variety of projects to transform urban subsystems (energy, waste, transport, blue-green ecological network), each of which requires analysis, decision-making, practical planning, and actual construction. It is conceivable that certain parts of the city development analysis in the first step are deferred to the more incremental, piecewise analysis in the second step. Particularly in SSP, the agenda readily sets outlines for activities for stakeholders in planning change. In UTM, the agenda outlines radical changes formulated by change agents, but leaves room for experimentation and learning, and thus needs to facilitate processes to do so. So, arguably, the toolkit for urban transition requires instruments for participatory vision formulation, analysis of the impacts of changes, co-creation of individual transformation projects involving different sets of stakeholders, and, ultimately, specification and execution of plans with room for experimentation and learning. This section



takes stock of the current toolkit of instruments for these process steps, specifically aiming to distinguish "types" of instruments based on their properties.

3.1. Three-Dimensional Spectrum for Instrument Classification

The selection and subsequent classification of instruments should be based on their (potential) use throughout the urban transition process, particularly in co-creating (the vision and plans for) projects and their implementation. Notable efforts have been made to map instruments to the various phases in the (short-term) urban planning process. Tasks and responsibilities in the planning and development process are often structured into consecutive phases (see e.g., Hopkins, 2001, p. 191), in each of which different instruments are used (see e.g., Siems, 2023, p. 53). Generally, a planning process consists of the following phases: (a) system analysis (using instruments for information and data collection, statistical analysis, and mapping), (b) collaborative goal setting (using instruments for brainstorming, preference analysis, and dealing with trade-offs between interests), (c) making action plans (e.g., using what-if analysis, cost-benefit analysis, roadmapping), and (4) project implementation and management. For long-term transition processes consisting of a multitude of subsystem transformation projects, activities can generally be separated into two broad phases: a phase of vision formulation and strategic planning followed by a phase of implementing a series of projects. In UTM, these projects take the form of experiments with monitoring, reflection, and provisions for adaptation. As such, one of the key dimensions of interest is whether instruments are of use in short-term urban transformation projects or rather support long-term planning, experimentation, or city development concerned with multiple short-term projects. Note that whether tools are applied over a longer time horizon or not has implications for the design and maintenance of these instruments, as well as for procedures and practices for updates based on intermediate developments.

In addition, there is ample literature on developments in instruments for two more dimensions. Particularly noteworthy is the increase in the use of digital tools (see e.g., Lieven et al., 2021; Staffans et al., 2020). This trend highlights the relevance of positioning instruments along a methodological-analytical dimension, ranging from low-tech tools for collecting, exchanging, and analyzing qualitative information and stakeholder perspectives to high-tech tools for quantitative, data-rich analysis, simulation, or optimization.

Moreover, planning instruments changed substantially during the "communicative turn," when stakeholder engagement, participation, and other governance modes came into focus (Faludi, 1973; Healey, 2020). Thus, the third dimension pertains to the governance mode of instruments, which ranges from being applied by experts for top-down decisions (as in rational-comprehensive planning, for example) to involving a variety of stakeholders such as urban planners, municipal governments, residents, utility companies, and construction firms in co-creation. There is also bottom-up, grassroots-driven change (e.g., tactical urbanism, transition towns), but strictly speaking, this is not an instrument that can be actively wielded. However, participants in such grassroots movements could be invited to join the transition arena. Moreover, transition planners may cultivate such movements and adopt successful initiatives for wider co-creative application.

In conclusion, instruments for co-creating urban transition can be positioned within a three-dimensional spectrum: (a) from qualitative, conceptual, data-extensive, low-tech instruments to quantitative, data-intensive, high-tech instruments; (b) from hierarchical, expert-driven planning to collaborative planning with stakeholder engagement; and (c) from short-term, singular projects to long-term transition agendas.



3.2. Methodology for Instrument Identification, Characterization, and Typology Formulation

The first aim of this article is to formulate a typology of instruments that are either already used or suitable for use in the co-creation of the urban transition. However, finding and identifying these instruments is challenging because the conceptions of the transition process, the planning of urban transition projects, and the usage of co-creation instruments are all relatively recent and therefore possibly insufficiently reported. Consequently, instrument designations and terms used vary across disciplines and stakeholders. Moreover, instruments differ substantially in nature (ranging from digital tools to participation methods to governance frameworks), and many exist that have not yet been applied in urban transition planning and management. City planners, consulting agencies, applied research institutes, and other actors are engaged in developing new instruments, so the entire toolkit, its applications, and associated keywords continue to evolve. Consequently, what one is searching for changes with insights from what is being found.

Given these challenges, this article adopts a structured exploratory snowballing approach based on selected principles of qualitative content analysis, particularly iterative deduction-induction, content interpretation (for keyword selection), and progressive typology development (Kluge, 2000; Mayring, 2000). The approach consists of two stages: iterative search and consolidation. The search alternates between a deductive step, in which keywords and sources are used to identify new instruments, and an inductive step, in which content analysis of instrument descriptions expands the keywords and sources used next. The keywords include both designations of actual instruments as well as generic properties that describe their characteristics and applications. In conjunction with the expansion of the set of instruments and progress in understanding their properties and applications, the (conceptual basis for the) selection and typology of instruments is also gradually refined (see notably Kluge, 2000, on integrating typology discovery in qualitative content analysis). Ultimately, there is a consolidation of the types of instruments. The methodology for instrument search and type identification is depicted in Figure 1 and described below in detail.







The iterative search for instruments alternates between deductive and inductive steps. In the deductive step, the search is conducted using (combinations of) keywords associated with co-creation, governance methods, digital tools, conceptual approaches, etc., looking for instruments that are or may be used in urban transition. This is aimed at creating a broad, diverse inventory. Initially, the list of keywords for search is based on the prior literature study. The search commenced in several directions: one on communication with and participation of stakeholders (using keywords such as "participat*," "engag*"), one on digital tools (e.g., "digital planning," "digital platform," "GIS"), and one on governance forms (e.g., "governance," "expert," "grassroots"). In general, these keywords were combined with keywords on the transition itself (e.g., "sustainab*," "green," "environment*"), and spatial context (e.g., "urban," "city," "town," "metropol*," "region*"). One general and two academic literature search engines (Scopus, Google Scholar) were used. Initially, primarily academic peer-reviewed literature was searched, but over iterations, this got expanded with conference proceedings (notably for digital tools), gray literature (consulting reports, white papers), and websites of (networks of) municipalities, (applied) research institutes, etc. Each discovered instrument was recorded in a database for further in-depth inquiry.

In the inductive step, specific designations of instruments (e.g., "citizen participation," "living lab," "DIPAS") are used for in-depth search to add a description, references to academic literature, and links to applications to the records in the database (occasionally also found in gray literature, websites of research institutes or municipalities, and so on). Content analysis of descriptions of the instruments and applications occasionally results in new instruments to be added for further in-depth analysis but also alternative keywords or new sources. In-depth search may also lead to merging records of highly similar instruments. As part of the inductive step, specific values were assigned to various instrument properties, such as Arnstein's ladder of participation, actors involved (the organizer and main participants), frequency of use, time interval of expected impact, and the direction, purpose, and type of information exchange. For most of these properties, serious attempts were made to characterize all instruments so that these properties could be used in devising a typology. After each inductive step of in-depth analysis of additional instruments, the authors discussed properties of instruments (and whether to add new properties for potential use in the typology later), instrument in- and exclusion criteria, and revision of inductive search directions, sources, and keywords.

This iterative process culminated in the identification and characterization of a wide variety of instruments, ranging from generic communication and voting tools, to early-stage participatory vision formulation and advanced decision-support tools for urban subsystem design and project planning. The iterative search stopped when newly identified instruments were mere variations or applications of those already in the dataset (cf. Saunders et al., 2018). At that point, the aim of finding the main *types* of instruments was presumably attained.

The subsequent consolidation stage focused on finalizing the instrument typology by first filtering instruments and then selecting properties for clustering into types. After the search stage, the dataset still contained numerous instruments recorded for further in-depth analysis, but for which limited evidence of active use for transition planning or management was found (e.g., "citizen science," "hackathon"). The filtering kept only instruments that can be used for formulating visions (e.g., "town hall meetings"), analysis of city developments (e.g., "using models," "expert panels"), setting actionable goals and planning tasks (e.g., "backcasting," "roadmapping"), or running projects (e.g., "living labs"). Since properties added during the iterative phase were not equally relevant, a subset was selected for clustering instruments into types. Several different subsets of properties were already considered for clustering during the iterative



stage. However, ultimately, the three key dimensions discussed in the previous section emerged as the basis for clustering, with one based on theory (time horizon) and the other two reflecting observed trends in instrument development (participation, digitalization).

3.3. Types of Instruments

The inventory of instruments led to the following four distinct types in the three-dimensional spectrum (see Figure 2).

3.3.1. Participatory Planning and Communication Platforms

The most generic type is that of communication and participation instruments used for information exchange, soliciting feedback, and collaboration in the formulation of strategies and plans without using computer models, advanced data-analytical means, or any need for advanced digital skills or expert domain knowledge. At its core, urban planning often has a regulatory role in which a planning or zoning commission (typically composed of city council members, officials, and sometimes advisors) reviews building proposals to ensure alignment with zoning, environmental guidelines, and community objectives as outlined in master plans and development strategies. Formulation and regulation used to be the responsibility of urban city planners and experts. However, ever since the "communicative turn," there has been a more explicit focus on collaboration and participation in urban planning (Healey, 2020). Prompted by actors seeking to change land use, by developments such as demographic or economic growth or decline, or following a broader call for mitigation and adaptation to climate change, urban planners may involve a wider set of stakeholders in the formulation of the urban development vision, strategies, regulation, and plans (see e.g., Hopkins, 2001). To this end, there is a large toolkit of participation instruments particularly suitable for making an inventory of transition challenges or formulating visions at the outset (e.g., focus groups, surveys, town hall meetings, citizens' assemblies, deliberative mini-publics, citizen councils/panels, planning cells, and community forums). On a variety of such participation formats, see Nanz and Fritsche (2012) and OECD (2020). Already actively used in later stages of urban redevelopment projects, city planners also have a toolkit for engagement with the general public and public communication tools (e.g., interactive exhibits, billboards, outdoor photo exhibits, scaffolding scrims). More recently, tools for online or offline digital participation (with labels such as e-participation or digital democracy) have emerged that seek to engage a larger and possibly broader group of citizens in policymaking and co-creation of public services (Ragi Eis Mendonca & De la Llata, 2023; Rodriguez Müller, 2022). Also for participation in urban planning, there are various online platforms (Abdalla et al., 2016; Giannoumis & Joneja, 2022; Hovik et al., 2022), offline digital tools, and apps (Ertiö, 2015; Falco & Kleinhans, 2018). There are numerous examples of how digital city platforms are used in participatory urban development and design (see e.g., Galassi et al., 2021; Giannoumis & Joneja, 2022; Lieven et al., 2021; Noennig et al., 2023). Given the novelty of such tools, the formulation of guidelines for online participation in urban planning is due (Afzalan & Muller, 2018; Hofmann et al., 2020).

Generally, participation instruments differ in the degree of engagement and actual decision-making power of participants (Arnstein, 1969). Some of the tools seem to be more suitable for consultation than genuine co-creation processes with community members and other stakeholders. In contrast, UTM focuses on instruments for co-creation with a deliberately picked set of *change agents* (e.g., roadmapping, formulation of a transition vision and agenda in the transition arena, collaborative policy formulation in a policy lab). Central



in transition management is also the element of experimentation and learning in a real-world setting which forms the basis for the second type of co-creation tool.

3.3.2. Urban Living Labs

There are several instruments for medium-term, real-world experimentation and co-creation, such as Reallabor (Schäpke et al., 2017), Urban Living Lab (Aernouts et al., 2023; Bulkeley et al., 2016; Mahmoud et al., 2021; Voytenko et al., 2016), Urban Transition Lab (Nevens et al., 2013), and Community Design Charrettes (Lennertz & Lutzenhiser, 2017; Roggema, 2014). We will use the term "urban living lab" as the general term to refer to this kind of instrument. Such urban living labs are concerned with ongoing participation, experimentation, and learning to find and test local solutions to societal challenges in real-life settings (Bulkeley et al., 2016; Nesti, 2018; Voytenko et al., 2016). However, urban living labs differ substantially in their goals and forms of the co-creation process (Nesti, 2018; Puerari et al., 2018). Moreover, some initiatives, such as transition towns (Connors & McDonald, 2011), are arguably a form of living lab, though arising from a grassroots movement or a bottom-up community-led or corporate frontrunner-led initiative, rather than a top-down initiative of researchers or urban planners. Given the open concept of urban living labs and the extended time interval over which such labs may persist, participants may also use other types of instruments (see e.g., the mapping of instruments to elements of the urban transition lab in Nevens et al., 2013, p. 116). This gives rise to hybrid forms, such as living labs using digital co-creation tools (Lieven et al., 2021; Mačiulienė & Skaržauskienė, 2020). However, practically, whenever such local, experimental, and temporary tactical urban development initiatives are considered successful, further diffusion and implementation would require at some point in time more expert-driven urban planning, adjustments to regulations, and actual construction project management. So, experimental co-creation leads the way for professional urban planning, which is rather the next type of instrument.

3.3.3. Expert Planning Support Systems

There is a toolkit used by urban planning professionals for short- to mid-term urban planning and design. The more traditional instruments of land use, master planning, and zoning ordinances are still used, albeit often also modernized. Planners use not only generic digital tools (e.g., QGIS, SketchUp, digital land use maps), but also data-intensive digital tools and systems for planning support (Geertman & Stillwell, 2004, 2009). Over the last few decades, these tools have advanced significantly, driven by developments in technology, data science, and the increasing availability of (real-time) data. A prominent advancement is the integration of GIS-based digital twins in tools for transportation and transit planning (e.g., TransCAD, TBEST, Hastus, PTV Visum), risk and resilience assessment (e.g., flood, heat, or disaster scenarios), and urban development simulations (e.g., UrbanSim, ArcGIS CityEngine). Modeling the urban subsystems typically requires the input of trends, scenarios, or models of demographic development, travel behavior, citizen preferences, etc. (see e.g., Landis, 2012), using, for example, agent-based models (Huynh et al., 2015; Namazi-Rad et al., 2017). Such tools support decisions on land use, locations for amenities, provision of public services, etc. Planners also have an interest in a digital twin model that represents the actual state of a focal city, sometimes even showing real-time data, such as on mobility patterns, energy consumption, etc. (see e.g., Anttiroiko, 2021; Gil, 2020). Such digital twins may even use real-time data collected in "smart cities" for operational management (e.g., of traffic flows) and computational urban planning (Douay, 2018; Geertman et al., 2019; Goodspeed et al., 2023). Such support tools can be used to identify challenges (e.g., traffic bottlenecks, heat islands), opportunities



(e.g., solar irradiance), and evaluate *incremental* changes (e.g., locations of schools, small transport network changes). These features make such tools suited for communication and collaborative planning with residents, firms, and other stakeholders. That said, despite their sophistication, these tools are inherently limited when it comes to planning for long-term, transformative urban change. After all, historical or present-day data on current conditions such as mobility patterns and energy usage have limited value in projections about the city in radical transformation scenarios. Therefore, these digital twin-based support tools are suitable for studying and planning *incremental* change, but not for visionary, radical transformation. This calls for the fourth type of instrument.

3.3.4. Virtual Transformation Labs

These instruments are virtual or physical laboratories for stakeholder involvement, collaborative planning, and decision-making using digital or hybrid physical-digital city models for long-term transition planning. The dividing line between the planning support tools discussed before and the transformation laboratories for virtual city development lies in the latter's persistent focus on multi-stakeholder interaction for strategic decisions and the ability to conduct radical, virtual experiments and scenario analysis over the long term. Virtual transformation labs use a flexibly configurable digital city development model to examine the impact of changes to the city (e.g., built environment, transportation network, green-blue infrastructure), notably using simulation, with the explicit intention to evaluate deliberate, radical changes, conduct "what-if" experimentation, and support formulation of visions and strategic planning. Examples of the latter are CityScope (by MIT; Alonso et al., 2018), Cockpit for Collaborative Urban Planning (HafenCity Hamburg, Austrian Institute of Technology, MIT Media Lab), Smart Region Lab (Hochschule Luzern), and the Decision



Figure 2. The four types in the three-dimensional spectrum of transition instruments.



Theater (Arizona State University). Although the technology and participation method are generic, the geographical scope, problem and system definitions, as well as building the network of change agents involved are time- and resource-intensive, so consequently mostly place-specific models and labs emerge. In these examples, the research institute, the city development model, and the main visualization platform are closely associated. Moreover, the content of the models, the interaction and interface of the visualization platform, and the use in and content of workshops are typically designed in participation with stakeholders. In the cases studied, the laboratories and underlying city models are not finished but continue to be developed for specific (applied) research projects. The development and application of these tools are therefore part of "action research." In real-world living labs, the progressive application and scaling up of experiments must be approached prudently, subject to monitoring and reflection, and with options for roll-back. In virtual laboratories, in contrast, experiments can be large-scale and radical, though their impacts are, of course, based largely on assumptions and simulations rather than observed outcomes.

3.4. Implementation Challenges of Digital Instruments

The descriptions of the four types of instruments contain indications on how they can be used in the co-creation of transition. However, the use of *digital planning support tools* (Type 3 in Figure 2) and *virtual transformation labs* (Type 4) is relatively recent. Despite the arguments in favor of the use of digital co-creation instruments, the actual implementation and application face several challenges, four of which will be discussed here.

First, there is the challenge of operationalizing the transition process. The transition is a long-term process with open-ended experimentation and complex interactions between subsystems to be transformed. Providing quantitative data and defining meaningful scenarios for analysis with digital city models is challenging, but without this, the tool has limited value. This is reminiscent of the so-called "implementation gap," i.e., the limited uptake of planning support systems despite their availability (Geertman & Stillwell, 2004, 2009), partly due to limited usefulness for strategic planning (te Brömmelstroet, 2017).

Second, there is the challenge of ensuring that the city transition model is location-specific and up-to-date. Further application is hindered by the substantial differences among cities in their transition challenges, urban subsystems, stakeholders involved, and future city visions. So, there is no one-size-fits-all template for the transition dynamics. Instead, tools, data collected, policy scenarios, and even transition management capabilities need to be tailored to such place-specificities (cf. Larbi et al., 2021). Moreover, the models and underlying data need to be updated frequently to reflect real-world developments. In addition, the city model and simulation scenarios need to be extended for new transition projects when they arise.

Third, planners are dependent on the developers to update the tools and notably underlying models. However, not uncommonly, the development of digital tools is part of academic action research of individual researchers or even research institutes. In fact, researcher-developers may well engage with the stakeholders for the specification of the digital city model and scenarios, participatory design of the interface and the tool's use cases, and have actual workshops for decision support as the final deliverable of the project. These tools may fail to meet the practical needs of planners. Moreover, for such tools, there is often no explicit intention to develop them commercially or maintain them in the future. Notable exceptions include digital participatory planning systems and laboratories associated with city or regional development institutes. Hereby, a base



platform is developed over time, with smaller tools for specific research purposes built on top. However, since these need to be made location-specific, one faces the second challenge.

Fourth, using digital co-creation tools also requires a change in the role and skills of urban planners. By design, engagement with stakeholders and a focus on mid- to long-term transformation are central. On top of their regular domain knowledge, planners would need to be able to operationally define city development models, specify transition scenarios, develop skills to work with simulation software, and analyze output data. Moreover, effective use of digital tools in planning workshops requires facilitators to make thoughtful interventions to ensure appropriate tool use and balance this with contextual discussions (Pelzer et al., 2015).

Urban planners face challenges not only in implementing and using these digital instruments, but also in adopting the mindset, procedures, and practices for transition co-creation. This is discussed in more detail in the next section.

4. Institutionalization of Co-Creation of the Urban Transition

The urban transition is a long-term process in which a portfolio of relatively short-term projects transforms various urban subsystems. The previous section identified instruments planners may use for the transition planning and management. In line with observations in SSP and, to a lesser extent, UTM, this transition process may also require procedural changes to planning practices. After all, instruments may need to be updated, projects need to be monitored, learning is to be facilitated, and the project portfolio is to be managed. Surprisingly, the literature reports little on the actual institutionalization of such transition planning and management. In fact, while UTM emphasizes that the transition agenda should be anchored, e.g., by engaging and mobilizing stakeholder networks around projects (Roorda et al., 2014; Wittmayer et al., 2018), such durable anchoring is rarely reported (Hölscher, 2018, p. 385). This section discusses four activities of transition management that need to be institutionalized: (a) building capabilities and activities for stakeholder engagement, governance, and co-creation; (b) developing and updating data, tools, and city models for analysis in subsystem transformation projects; (c) building capability for and facilitating learning across projects at the city level; and (d) learning from similar projects as well as transition co-creation methods in other cities. The capabilities and their relationships are depicted in Figure 3 and explained in detail below.

4.1. Building Capabilities for Stakeholder Engagement, Governance, and Co-Creation

Both SSP and UTM advocate a change in planning praxis toward engagement with and collaboration between a diverse set of stakeholders far beyond the zoning committee. Given that the urban transition is a unique undertaking for each city, urban planners and the various stakeholders involved will (have to) develop capabilities, both for co-creation of the top-level vision, strategy, and agenda with change agents as well as for subsystem transformation projects (cf. Albrechts, 2013; Frantzeskaki et al., 2018) involving stakeholders closer to the actual construction (e.g., system experts, construction companies, affected citizens, relevant municipal departments). In both approaches, co-creation requires alternative governance structures and long-term commitment, and therefore institutionalization (see e.g., Mahmoud et al., 2021). This shift toward co-creation calls for new skills and education for urban planners (Rooij & Frank, 2016).



4.4 Monitoring, learning, adopting practices from elsewhere (e.g. city networks)





Figure 3. Schematic overview of capabilities in managing the various urban transition projects. Arrows indicate information flow between different activities for the purpose indicated by the labels. Descriptions are provided in the sections with the given numbers.

Planners must develop capabilities to navigate the different perspectives and interests of the participants, and cope with conflicts and trade-offs by creating effective compromises (cf. Albrechts, 2013).

In SSP, co-creation is to be integrated into the day-to-day planning praxis for both vision formulation as well as proactive co-creation and alignment of real-world transition projects. In this case, co-creation is institutionalized and integrated into urban planning and design, regulations for public utilities, and so on. In UTM, the transition arena is deliberately placed outside of the regular planning institutions (Wolfram, 2018, p. 111). In that case, urban planners need new skills and use new tools in the transition team with a set of *change agents*. Both in UTM and SSP, they need to build governance capabilities used within individual projects (cf. Molenaar et al., 2021). An extensive operationalization of transformative capabilities, notably with regard to governance, engagement, and co-creation, is found in Shahani et al. (2022).

4.2. Development, Participatory Design, and Updating of (Digital) Instruments and City Development Model

As seen in Section 3.4, using transition planning instruments has various challenges and requires planning offices to acquire new capabilities. Notably, digital instruments change planning practices substantially (see e.g., Nummi et al., 2023) and planners need new digital skills to use them (Sabri & Witte, 2023). Moreover, these instruments are typically not developed by the municipal planning office, but rather by a software



company and occasionally by an applied research institute. To ensure planners use these instruments ("closing the implementation gap"), they need to be actively involved in their design. Moreover, the actual implementation requires the active contributions of public and private sector stakeholders, decision-makers, and local construction companies in model design to ensure correct representation of urban subsystems and to provide data used in the digital city model. Each transition project in the portfolio (e.g., mobility, energy, waste) requires stakeholder engagement in the formulation of subsystem operationalizations and development scenarios, as well as participatory design of the digital twin or development model, and analysis of experimental or simulation results for decision support. Moreover, external events, renewed insights, as well as the need to analyze previously unforeseen transition projects may call for changes to the digital development model. To remain relevant and useful, the planners need to update the data used and, if necessary, modules in the city development model pertaining to the subsystem transformed. Following such updates, new computational results need to be produced and analyzed. The urban planning office needs to build capabilities for these recurring activities.

4.3. Building Capabilities for Management of and Learning Across the Portfolio of Transition Projects

As emphasized, the long-term urban transition comprises an evolving portfolio of relatively short-term projects transforming urban subsystems. Particularly in the SSP line of reasoning, the urban planning office would hold responsibility for managing this portfolio by initiating projects top-down as well as subsuming interesting ones that are initiated bottom-up (e.g., by local communities, non-governmental organizations, or entrepreneurs). The planning office would need to build legal, financial, and organizational capabilities to initiate, adopt, and develop projects, align these projects with the long-term vision, engage stakeholders in practical implementation, and analyze and deal with interactions between projects.

Given the multitude of projects, the focus of learning for planners is not just *within* individual projects but also *across* both different ones in the same city as well as similar ones elsewhere (See Section 4.4). It becomes relevant to adopt and co-develop methods, instruments, and city development models that can be used in the various projects and possibly analyze the interaction of these projects.

In addition, with the substantial chance of both exogenous changes (e.g., regulations or policies at higher spatial scales, changes in national infrastructures) as well as the emergence of unforeseen endogenous challenges, the long-term planning process is generally not linear but rather iterative, requiring revisions and adaptations. Arguably, developments in projects may force planners to revise the vision or the "radical" transition agenda, which in turn may feed-forward in projects.

4.4. Learning in City Networks at Project Level, Planning Capabilities, and Transition Praxis

Cities are the niches for experimental urban subsystem transformation (cf. Torrens et al., 2019). Some transition projects may be suitable for small-scale experiments that can be rolled out and scaled up. Other projects require costly, near-irreversible, large-scale implementations that do not allow experimentation. In this case, one could use simulation with digital models, for example. Learning about experiments and projects conducted elsewhere may not only spark ideas for new local projects but also provide clues about effective implementations of experiments or projects readily pursued.



In a network of transition cities (e.g., C40, EU 100 cities, Global Covenant of Mayors), urban planners, change agents, and other stakeholders may learn from processes, experiments, and project results in other cities. Clearly, due to local particularities, "best practices" cannot be directly applied to other cities per se (Bulkeley, 2006), but must be tailored to new conditions. Moreover, apart from learning about ideas, particularities, tools, and methods for individual projects, one may also seek insights into planning capabilities, instrumentation, as well as digital city development models for urban multi-system analysis. Note that information shared in such city networks would allow comparative multi-city studies and thus contribute to the development of methods, learning on the role of local particularities, and ultimately the advancement of urban transition theory (Roorda et al., 2014; Webb et al., 2018; Wittmayer et al., 2016).

5. Conclusions and Discussion

Recently, urban transition and planning theory scholars turned their attention to the co-creation of the radical transformation of urban subsystems. However, it is not clear what co-creation is in practice, which instruments exist, and how these instruments can be used effectively. This is partly caused by the ambiguous and rapidly developing conceptions of the urban transition process. Based on SSP and UTM, the urban transition is conceived as a long-term process driven by a co-created vision and agenda with recurring management of an evolving portfolio of relatively short-term co-creation projects to transform urban subsystems (e.g., infrastructures, utilities, services, built environment). This article identifies types of instruments and examines the institutionalization of capabilities for their application and development.

Based on the inventory of instruments, four types are identified (participatory planning and communication, expert planning support systems, urban living labs, and virtual transformation labs) that are particularly suitable either for conceptual exploration or computational analysis, and either for short planning horizons or long-term, iterative co-creation. The co-creation tools and the living labs offer platforms for engagement and vision formulation, whereas the digital tools offer valuable complementary ways to analyze subsystem transformations. Particularly the "virtual transformation lab" is well suited to using city *development* models for strategic decision-making. Given that the three dimensions for classification as well as the clustering of instruments are based on qualitative content analysis rather than theoretical arguments, follow-up research should take a critical look at the typology and instrument classifications.

Given that transition challenges evolve, regular updates to models and data, alignment with planning needs, and continuous stakeholder engagement are essential. In addition, urban transition planners need to engage stakeholders, manage project portfolios, and foster continuous learning within and across projects. This underlines how co-creation with these instruments should be institutionalized within transition planning and management practices. Further case studies are needed to gain insight into the challenges of developing and applying such instruments, as well as building the required capabilities.

The operational and managerial activities thus discerned could be performed by a "Transition Planning Office" (TPO). This TPO could host the capabilities for stakeholder engagement in collaborative planning, participatory design of models, collection of city data, etc. It could manage the portfolio of projects, initiate or subsume projects, provide legal and financial support for individual projects, align projects with the overarching vision or resolve conflicts and exploit synergies between projects. It may deal with uncertainty in strategic planning by making substantive learning within and across projects integral to procedures and the planning mindset.



Moreover, it could build capabilities for all four types of instruments, e.g., contribute to an urban living lab or co-develop a city model for a virtual transformation lab. Additional research is needed to define such a TPO and assess its viability in planning practices. The notion of a TPO extends the conceptual frameworks used. It complements UTM's temporary stand-alone transition arena with active involvement in long-term practical planning and implementation. In addition, it complements SSP's call to make urban planning more strategic by building capabilities for operational planning, managing a portfolio of projects, and developing and using a variety of instruments. Arguably, a limitation of this article is its exclusive focus on SSP and UTM. Further research should explore alternative perspectives on planning (e.g., adaptive urban planning) and urban change (e.g., tactical urbanism), which may offer different conceptions of the transition process, stakeholder participation, or the use of instruments.

Further research is required to examine the place-specificities of transition challenges and the implications for instrumentation and institutionalization. After all, transition challenges and planning practices in megacities differ from those in towns, and those in cities in developing countries differ from those in developed countries (cf. Parnell & Robinson, 2012; Watson, 2016), both in terms of existing and envisioned urban subsystems, infrastructural elements, socio-economic factors, governance capabilities and styles, data acquisition capabilities, and required institutional and instrument developments. In addition, given that urban transition initiatives are embedded in development policies and sustainability initiatives at regional, national, and international levels, greater multilevel alignment may be needed to create a supportive regulatory environment (cf. Bulkeley & Betsill, 2005).

Acknowledgments

The first author and second author gratefully acknowledge funding from the Federal Ministry of Education and Research (Bundesministerium für Bildung und Forschung [BMBF]) for the project hyBit: Hydrogen for Bremen's Industrial Transformation (Grant 03SF0687A). The first author also acknowledges support from the European Union Horizon Europe programme through Grant Agreement 101132559 (ST4TE: Strategies for Just and Equitable Transitions in Europe). Additionally, the second author gratefully acknowledges funding from the Federal Ministry of Education and Research (Bundesministerium für Bildung und Forschung [BMBF]) for the project Quarree100: Resilient, Integrated, and System-Serving Energy Supply Systems in Existing Urban Districts With Full Integration of Renewable Energies (Grant 03SBE113B).

Funding

Publication of this article in open access was made possible through the institutional membership agreement between Maastricht University and Cogitatio Press.

Conflict of Interests

The authors declare no conflict of interests.

References

- Abdalla, S. S., Elariane, S. A., & El Defrawi, S. H. (2016). Decision-making tool for participatory urban planning and development: Residents' preferences of their built environment. *Journal of Urban Planning and Development*, 142(1), Article 04015011.
- Aernouts, N., Cognetti, F., & Maranghi, E. (Eds.). (2023). Urban living lab for local regeneration: Beyond participation in large-scale social housing estates. Springer. https://doi.org/10.1007/978-3-031-19748-2



- Afzalan, N., & Muller, B. (2018). Online participatory technologies: Opportunities and challenges for enriching participatory planning. *Journal of the American Planning Association*, 84(2), 162–177. https://doi.org/ 10.1080/01944363.2018.1434010
- Albrechts, L. (2004). Strategic (spatial) planning reexamined. *Environment and Planning B: Planning and Design*, 31(5), 743–758. https://doi.org/10.1068/b3065
- Albrechts, L. (2013). Reframing strategic spatial planning by using a coproduction perspective. *Planning Theory*, 12(1), 46–63. https://doi.org/10.1177/1473095212452722
- Albrechts, L. (2015). Ingredients for a more radical strategic spatial planning. *Environment and Planning B: Planning and Design*, 42(3), 510–525. https://doi.org/10.1068/b130104p
- Allmendinger, P., & Tewdwr-Jones, M. (Eds.). (2001). *Planning futures: New directions for planning theory*. Routledge. https://doi.org/10.4324/9780203996195
- Alonso, L., Zhang, Y. R., Grignard, A., Noyman, A., Sakai, Y., ElKatsha, M., Doorley, R., & Larson, K. (2018). CityScope: A data-driven interactive simulation tool for urban design. Use case Volpe. In A. J. Morales, C. Gershenson, D. Braha, A. A. Minai, & Y. Bar-Yam (Eds.), *Unifying themes in complex systems IX* (pp. 253–261). Springer. https://doi.org/10.1007/978-3-319-96661-8_27
- Anttiroiko, A.-V. (2021). Digital urban planning platforms: The interplay of digital and local embeddedness in urban planning. *International Journal of E-Planning Research*, 10(3), 35–49. https://doi.org/10.4018/IJEPR. 20210701.oa3
- Arnstein, S. R. (1969). A ladder of citizen participation. *Journal of the American Institute of Planners*, 35(4), 216–224. https://doi.org/10.1080/01944366908977225
- Bai, X., Baynes, T. M., Webb, R., Ryan, C., & Smith, M. H. (2021). Cities. In K. G. H. Baldwin, M. Howden, M. H. Smith, K. Hussey, & P. J. Dawson (Eds.), *Transitioning to a prosperous, resilient and carbon-free economy: A guide for decision-makers* (pp. 271–300). Cambridge University Press. https://doi.org/10.1017/9781316389553.016
- Batty, M. (2001). Models in planning: Technological imperatives and changing roles. *International Journal of Applied Earth Observation and Geoinformation*, *3*(3), 252–266. https://doi.org/10.1016/S0303-2434(01) 85032-7
- Brandsen, T., & Honingh, M. (2018). Definitions of co-production and co-creation. In T. Brandsen, T. Steen, &B. Verschuere (Eds.), *Co-production and co-creation* (1st ed., pp. 9–17). Routledge.
- Bulkeley, H. (2006). Urban sustainability: Learning from best practice? *Environment and Planning A: Economy and Space*, 38(6), 1029–1044. https://doi.org/10.1068/a37300
- Bulkeley, H., & Betsill, M. (2005). Rethinking sustainable cities: Multilevel governance and the "urban" politics of climate change. *Environmental Politics*, 14(1), 42–63. https://doi.org/10.1080/0964401042000310178
- Bulkeley, H., Coenen, L., Frantzeskaki, N., Hartmann, C., Kronsell, A., Mai, L., Marvin, S., McCormick, K., Van Steenbergen, F., & Voytenko Palgan, Y. (2016). Urban living labs: Governing urban sustainability transitions. *Current Opinion in Environmental Sustainability*, *22*, 13–17. https://doi.org/10.1016/j.cosust. 2017.02.003
- Connors, P., & McDonald, P. (2011). Transitioning communities: Community, participation and the Transition Town movement. *Community Development Journal*, 46(4), 558–572. https://doi.org/10.1093/cdj/bsq014
 Douay, N. (2018). Urban planning in the digital age. Wiley.
- Ertiö, T.-P. (2015). Participatory apps for urban planning—Space for improvement. *Planning Practice & Research*, 30(3), 303–321. https://doi.org/10.1080/02697459.2015.1052942
- Falco, E., & Kleinhans, R. (2018). Digital participatory platforms for co-production in urban development: A systematic review. *International Journal of E-Planning Research*, 7(3), 52–79. https://doi.org/10.4018/ IJEPR.2018070105



Faludi, A. (1973). A reader in planning theory (1st ed.). Pergamon Press.

- Frantzeskaki, N. (2022). Bringing transition management to cities: Building skills for transformative urban governance. *Sustainability*, 14(2), Article 650. https://doi.org/10.3390/su14020650
- Frantzeskaki, N., Hölscher, K., Bach, M., & Avelino, F. (Eds.). (2018). *Co-creating sustainable urban futures:* A primer on applying transition management in cities. Springer. https://doi.org/10.1007/978-3-319-69273-9
- Fugini, M., Bracci, E., & Sicilia, M. (Eds.). (2016). Co-production in the public sector. Springer. https://doi.org/ 10.1007/978-3-319-30558-5
- Galassi, A., Petríková, L., & Scacchi, M. (2021). Digital technologies for community engagement in decision-making and planning process. In A. Bisello, D. Vettorato, H. Haarstad, & J. Borsboom-van Beurden (Eds.), Smart and sustainable planning for cities and regions (pp. 387–397). Springer. https://doi.org/10.1007/ 978-3-030-57332-4_27
- Garde, A. (2020). New urbanism: Past, present, and future. Urban Planning, 5(4), 453–463. https://doi.org/ 10.17645/up.v5i4.3478
- Geertman, S., & Stillwell, J. (2004). Planning support systems: An inventory of current practice. *Computers, Environment and Urban Systems,* 28(4), 291–310. https://doi.org/10.1016/S0198-9715(03)00024-3
- Geertman, S., & Stillwell, J. (Eds.). (2009). Planning support systems: Best practice and new methods. Springer. https://doi.org/10.1007/978-1-4020-8952-7
- Geertman, S., Zhan, Q., Allan, A., & Pettit, C. (2019). *Computational urban planning and management for smart cities*. Springer.
- Giannoumis, G. A., & Joneja, N. (2022). Citizen participation and ICT for urban development in Oslo. In S. Hovik, G. A. Giannoumis, K. Reichborn-Kjennerud, J. M. Ruano, I. McShane, & S. Legard (Eds.), *Citizen participation in the information society: Comparing participatory channels in urban development* (pp. 97–116). Springer. https://doi.org/10.1007/978-3-030-99940-7_5
- Gil, J. (2020). City information modelling: A conceptual framework for research and practice in digital urban planning. *Built Environment*, 46(4), 501–527. https://doi.org/10.2148/BENV.46.4.501
- Goodspeed, R., Sengupta, R., Kyttä, M., & Pettit, C. (Eds.). (2023). Intelligence for future cities: Planning through big data and urban analytics. Springer Nature. https://doi.org/10.1007/978-3-031-31746-0
- Healey, P. (2020). Collaborative planning: Shaping places in fragmented societies. Bloomsbury Publishing.
- Hofmann, M., Münster, S., & Noennig, J. R. (2020). A theoretical framework for the evaluation of massive digital participation systems in urban planning. *Journal of Geovisualization and Spatial Analysis*, 4(1), Article 3. https://doi.org/10.1007/s41651-019-0040-3
- Hölscher, K. (2018). So what? Transition management as a transformative approach to support governance capacities in cities. In N. Frantzeskaki, K. Hölscher, M. Bach, & F. Avelino (Eds.), *Cocreating sustainable urban futures: A primer on applying transition management in cities* (Vol. 11, pp. 375–396). Springer. https://doi.org/ 10.1007/978-3-319-69273-9
- Hopkins, L. D. (2001). Urban development: The logic of making plans. Island Press.
- Hovik, S., Giannoumis, G. A., Reichborn-Kjennerud, K., Ruano, J. M., McShane, I., & Legard, S. (Eds.). (2022). *Citizen participation in the information society: Comparing participatory channels in urban development.* Springer. https://doi.org/10.1007/978-3-030-99940-7
- Huynh, N., Perez, P., Berryman, M., & Barthélemy, J. (2015). Simulating transport and land use interdependencies for strategic urban planning—An agent based modelling approach. Systems, 3(4), 177–210. https://doi.org/10.3390/systems3040177
- Jepson, E. J., Jr., & Haines, A. L. (2014). Zoning for sustainability: A review and analysis of the zoning ordinances



of 32 cities in the United States. Journal of the American Planning Association, 80(3), 239–252. https://doi.org/10.1080/01944363.2014.981200

- Keivani, R. (2009). A review of the main challenges to urban sustainability. *International Journal of Urban Sustainable Development*, 1(1/2), 5–16. https://doi.org/10.1080/19463131003704213
- Kluge, S. (2000). Empirically grounded construction of types and typologies in qualitative social research. Forum Qualitative Sozialforschung/Forum: Qualitative Social Research, 1(1), Article 14. https://doi.org/ 10.17169/fqs-1.1.1124
- Landis, J. D. (2012). Modeling urban systems. In R. Crane & R. Weber (Eds.), *The Oxford handbook of urban planning* (pp. 323–351). Oxford University Press. https://doi.org/10.1093/oxfordhb/9780195374995. 013.0017
- Larbi, M., Kellett, J., Palazzo, E., & Mehdipour, A. (2021). Urban sustainability transitions in two frontrunner cities: Insights from the multi-level perspective. *Planning Practice & Research*, 36(5), 494–513. https://doi. org/10.1080/02697459.2021.1919430
- Leino, H., & Puumala, E. (2021). What can co-creation do for the citizens? Applying co-creation for the promotion of participation in cities. *Environment and Planning C: Politics and Space*, 39(4), 781–799. https://doi.org/10.1177/2399654420957337
- Lennertz, B., & Lutzenhiser, A. (2017). The Charrette handbook: The essential guide to design-based public involvement (2nd ed.). Routledge. https://doi.org/10.4324/9781351179263
- Lieven, C., Lüders, B., Kulus, D., & Thoneick, R. (2021). Enabling digital co-creation in urban planning and development. In A. Zimmermann, R. J. Howlett, & L. C. Jain (Eds.), *Human centred intelligent systems* (pp. 415–430). Springer. https://doi.org/10.1007/978-981-15-5784-2_34
- Loorbach, D. A. (2022). Designing radical transitions: A plea for a new governance culture to empower deep transformative change. *City, Territory and Architecture*, 9(1), Article 30. https://doi.org/10.1186/s40410-022-00176-z
- Mačiulienė, M., & Skaržauskienė, A. (2020). Sustainable urban innovations: Digital co-creation in European living labs. *Kybernetes*, 49(7), 1969–1986. https://doi.org/10.1108/K-07-2019-0514
- Mahmoud, I. H., Morello, E., Ludlow, D., & Salvia, G. (2021). Co-creation pathways to inform shared governance of urban living labs in practice: Lessons from three European projects. *Frontiers in Sustainable Cities*, *3*, Article 690458. https://doi.org/10.3389/frsc.2021.690458
- Malekpour, S., Brown, R. R., & de Haan, F. J. (2015). Strategic planning of urban infrastructure for environmental sustainability: Understanding the past to intervene for the future. *Cities*, 46, 67–75. https://doi.org/10.1016/j.cities.2015.05.003
- Mayring, P. (2000). Qualitative content analysis. Forum Qualitative Sozialforschung/Forum: Qualitative Social Research, 1(2), Article 20. https://doi.org/10.17169/fqs-1.2.1089
- Molenaar, A., Hölscher, K., Loorbach, D., & Verlinde, J. (2021). Making the transition: Transformative governance capacities for a resilient Rotterdam. In R. de Graaf-van Dinther (Ed.), *Climate resilient urban areas: Governance, design and development in coastal delta cities* (pp. 175–190). Springer. https://doi.org/10.1007/978-3-030-57537-3_9
- Namazi-Rad, M.-R., Padgham, L., Perez, P., Nagel, K., & Bazzan, A. (Eds.). (2017). Agent based modelling of urban systems. Springer. https://doi.org/10.1007/978-3-319-51957-9
- Nanz, P., & Fritsche, M. (2012). Handbuch Bürgerbeteiligung: Verfahren und Akteure, Chancen und Grenzen. Bundeszentrale für Politische Bildung.
- Nesti, G. (2018). Co-production for innovation: The urban living lab experience. *Policy and Society*, 37(3), 310–325. https://doi.org/10.1080/14494035.2017.1374692



- Nevens, F., Frantzeskaki, N., Gorissen, L., & Loorbach, D. (2013). Urban transition labs: Co-creating transformative action for sustainable cities. *Journal of Cleaner Production*, 50, 111–122. https://doi.org/ 10.1016/j.jclepro.2012.12.001
- Noennig, J. R., Barski, J., Borgmann, K., & Baeza, J. L. (2023). Digital city science—A platform methodology for sustainable urban development. In P. Droege (Ed.), *Intelligent environments* (2nd ed., pp. 475–497). North-Holland. https://doi.org/10.1016/B978-0-12-820247-0.00013-8
- Nummi, P., Staffans, A., & Helenius, O. (2023). Digitalizing planning culture: A change towards information model-based planning in Finland. *Journal of Urban Management*, 12(1), 44–56. https://doi.org/10.1016/ j.jum.2022.12.001
- OECD. (2020). Innovative citizen participation and new democratic institutions: Catching the deliberative wave. https://www.oecd-ilibrary.org/governance/innovative-citizen-participation-and-new-democratic-institutions_339306da-en
- Parnell, S., & Robinson, J. (2012). (Re)theorizing cities from the Global South: Looking beyond neoliberalism. *Urban Geography*, 33(4), 593–617. https://doi.org/10.2747/0272-3638.33.4.593
- Pelzer, P., Goodspeed, R., & te Brömmelstroet, M. (2015). Facilitating PSS workshops: A conceptual framework and findings from interviews with facilitators. In S. Geertman, J. Ferreira, Jr., R. Goodspeed, & J. Stillwell (Eds.), *Planning support systems and smart cities* (pp. 355–369). Springer. https://doi.org/10.1007/978-3-319-18368-8_19
- Puerari, E., De Koning, J. I. J. C., Von Wirth, T., Karré, P. M., Mulder, I. J., & Loorbach, D. A. (2018). Co-creation dynamics in urban living labs. *Sustainability*, *10*(6), Article 1893. https://doi.org/10.3390/su10061893
- Ragi Eis Mendonca, B., & De la Llata, S. (2023). Toward digital democracy, e-planning and e-participation practices: The use and value of ICTs in the digital era. In A. Rubbo, J. Du, M. R. Thomsen, & M. Tamke (Eds.), *Design for resilient communities* (pp. 365–375). Springer.
- Raven, J., Stone, B., Mills, G., Towers, J., Katzschner, L., Leone, M. F., Gaborit, P., Georgescu, M., Hariri, M., Lee, J., LeJava, J., Sharifi, A., Visconti, C., & Rudd, A. (2018). Urban planning and urban design. In C. Rosenzweig, W. D. Solecki, P. Romero-Lankao, S. Mehrotra, S. Dhakal, & S. Ali Ibrahim (Eds.), *Climate change and cities: Second assessment report of the Urban Climate Change Research Network* (pp. 139–172). Cambridge University Press. https://doi.org/10.1017/9781316563878.012
- Rodriguez Müller, A. P. (2022). Citizens engagement in policy making: Insights from an e-participation platform in Leuven, Belgium. In T. Randma-Liiv & V. Lember (Eds.), *Engaging citizens in policy making: E-participation practices in Europe* (pp. 180–195). Edward Elgar Publishing. https://doi.org/10.4337/9781800374362. 00020
- Roggema, R. (Ed.). (2014). The design charrette: Ways to envision sustainable futures. Springer. https://doi.org/ 10.1007/978-94-007-7031-7
- Rooij, R., & Frank, A. I. (2016). Educating spatial planners for the age of co-creation: The need to risk community, science and practice involvement in planning programmes and curricula. *Planning Practice & Research*, 31(5), 473–485. https://doi.org/10.1080/02697459.2016.1222120
- Roorda, C., Wittmayer, J., Henneman, P., Van Steenbergen, F., Frantzeskaki, N., & Loorbach, D. (2014). *Transition management in the urban context: Guidance manual*. DRIFT, Erasmus University Rotterdam.
- Sabri, S., & Witte, P. (2023). Digital technologies in urban planning and urban management. *Journal of Urban Management*, 12(1), 1–3. https://doi.org/10.1016/j.jum.2023.02.003
- Saunders, B., Sim, J., Kingstone, T., Baker, S., Waterfield, J., Bartlam, B., Burroughs, H., & Jinks, C. (2018). Saturation in qualitative research: Exploring its conceptualization and operationalization. *Quality & Quantity*, *52*(4), 1893–1907. https://doi.org/10.1007/s11135-017-0574-8



- Schäpke, N., Stelzer, F., Bergmann, M., Singer-Brodowski, M., Wanner, M., Caniglia, G., & Lang, D. J. (2017). Reallabore im Kontext Transformativer Forschung: Ansatzpunkte zur Konzeption und Einbettung in den internationalen Forschungsstand (IETSR Discussion Papers in Transdisciplinary Sustainability Research No. 1/2017). Leuphana University of Lüneburg. http://rgdoi.net/10.13140/RG.2.2.28604.23687
- Seto, K. C., Davis, S. J., Mitchell, R. B., Stokes, E. C., Unruh, G., & Ürge-Vorsatz, D. (2016). Carbon lock-in: Types, causes, and policy implications. *Annual Review of Environment and Resources*, 41, 425–452. https://doi.org/10.1146/annurev-environ-110615-085934
- Shahani, F., Pineda-Pinto, M., & Frantzeskaki, N. (2022). Transformative low-carbon urban innovations: Operationalizing transformative capacity for urban planning. *Ambio*, *51*(5), 1179–1198. https://doi.org/ 10.1007/s13280-021-01653-4
- Siems, T. (2023). Stadt vermitteln. Birkhäuser. https://doi.org/10.1515/9783035624199
- Silva, C. N. (Ed.). (2010). Handbook of research on e-planning: ICTs for urban development and monitoring. IGI Global. https://doi.org/10.4018/978-1-61520-929-3
- Staffans, A., Kahila-Tani, M., & Kyttä, M. (2020). Participatory urban planning in the digital era. In S. Geertman & J. Stillwell (Eds.), *Handbook of planning support science* (pp. 307–322). Edward Elgar Publishing. https://www.elgaronline.com/display/edcoll/9781788971072/9781788971072.00030.xml
- Talen, E. (2012). Sustainability. In R. Crane & R. Weber (Eds.), *The Oxford handbook of urban planning* (pp. 120–140). Oxford University Press. https://doi.org/10.1093/oxfordhb/9780195374995.013.0007
- te Brömmelstroet, M. (2017). PSS are more user-friendly, but are they also increasingly useful? *Transportation Research Part A: Policy and Practice*, 104, 96–107. https://doi.org/10.1016/j.tra.2016.08.009
- Torrens, J., Schot, J., Raven, R., & Johnstone, P. (2019). Seedbeds, harbours, and battlegrounds: On the origins of favourable environments for urban experimentation with sustainability. *Environmental Innovation and Societal Transitions*, 31, 211–232. https://doi.org/10.1016/j.eist.2018.11.003
- van de Ven, F. H. M., Snep, R. P. H., Koole, S., Brolsma, R., van der Brugge, R., Spijker, J., & Vergroesen, T. (2016). Adaptation Planning Support Toolbox: Measurable performance information based tools for co-creation of resilient, ecosystem-based urban plans with urban designers, decision-makers and stakeholders. *Environmental Science & Policy*, 66, 427–436. https://doi.org/10.1016/j.envsci.2016.06.010
- Voytenko, Y., McCormick, K., Evans, J., & Schliwa, G. (2016). Urban living labs for sustainability and low carbon cities in Europe: Towards a research agenda. *Journal of Cleaner Production*, 123, 45–54. https://doi.org/ 10.1016/j.jclepro.2015.08.053
- Walsh, P. (2018). Translating transitions thinking and transition management into the city planning world. In N. Frantzeskaki, K. Hölscher, M. Bach, & F. Avelino (Eds.), *Co-creating sustainable urban futures: A primer on applying transition management in cities* (pp. 261–285). Springer. https://doi.org/10.1007/978-3-319-69273-9_11
- Watson, V. (2016). Shifting approaches to planning theory: Global North and South. Urban Planning, 1(4), 32–41. https://doi.org/10.17645/up.v1i4.727
- Webb, R., Bai, X., Smith, M. S., Costanza, R., Griggs, D., Moglia, M., Neuman, M., Newman, P., Newton, P., Norman, B., Ryan, C., Schandl, H., Steffen, W., Tapper, N., & Thomson, G. (2018). Sustainable urban systems: Co-design and framing for transformation. *Ambio*, 47(1), 57–77. https://doi.org/10.1007/s13280-017-0934-6
- Wittmayer, J. M., & Loorbach, D. (2016). Governing transitions in cities: Fostering alternative ideas, practices, and social relations through transition management. In D. Loorbach, J. M. Wittmayer, H. Shiroyama, J. Fujino, & S. Mizuguchi (Eds.), *Governance of urban sustainability transitions: European and Asian experiences* (pp. 13–32). Springer. https://doi.org/10.1007/978-4-431-55426-4_2



- Wittmayer, J. M., Mizuguchi, S., Rach, S., & Fujino, J. (2016). City networks for sustainability transitions in Europe and Japan. In D. Loorbach, J. M. Wittmayer, H. Shiroyama, J. Fujino, & S. Mizuguchi (Eds.), *Governance of urban sustainability transitions: European and Asian experiences* (pp. 33–45). Springer. https:// doi.org/10.1007/978-4-431-55426-4_3
- Wittmayer, J. M., van Steenbergen, F., Frantzeskaki, N., & Bach, M. (2018). Transition management: Guiding principles and applications. In N. Frantzeskaki, K. Hölscher, M. Bach, & F. Avelino (Eds.), Cocreating sustainable urban futures: A primer on applying transition management in cities (pp. 81–101). Springer. https:// doi.org/10.1007/978-3-319-69273-9_4
- Wolfram, M. (2018). Urban planning and transition management: Rationalities, instruments and dialectics. In N. Frantzeskaki, K. Hölscher, M. Bach, & F. Avelino (Eds.), *Cocreating sustainable urban futures: A primer on applying transition management in cities* (pp. 103–125). Springer. https://doi.org/10.1007/978-3-319-69273-9_5

About the Authors



Ben Vermeulen is an innovation economist researching regional green and digital transition processes using computer models, stakeholder engagement, governance, and policy interventions.



Lennart Winkeler researches agent-based models for socio-technical systems and supports a resilient energy transition through interactive decision tools. Drawing from experience in residential and industrial projects, he focuses on co-design processes and the integration of theory and practice to enhance energy transition outcomes.



Mohar Kalra is an artist and researcher exploring how we are entangled with the technological and ecological systems that undergird our built environments. He currently makes interactive public interventions that try to foster a more personal, vernacular knowledge of the natural and unnatural spaces that surround us everyday.